Author's response to reviews

Title: Primary pineal melanoma presenting in a patient with 22-year-old: a case report

Authors:

Parisa Azimi (parisa.azimi@gmail.com)
Hassan Reza Mohmmadi (mohammadi5@sbmu.ac.ir)
Mitra Refiezadeh (m.refiezadeh@yahoo.com)

Version: 2 Date: 12 October 2011

Author's response to reviews:

Adverse drug effects and drug Interactions was not found Pre and postoperative. Diagnoses of disease was delayed almost a week, after the hospitalization due to rare disease. No standard treatment guidelines exist for Primary pineal melanoma disease, but the deteriorating condition of our patient led to the decision to surgery. Not found any unexpected association between diseases or symptoms. Postoperatively, the patient did not saw new neurologic deficit and then unfortunately died from a cardiopulmonary arrest after 12 weeks since the initial symptoms. The cause of the disease and cardiopulmonary arrest were not clear to us.